1. Reviewer #1:

   **Specific Comments to Authors:** Dear authors. It results of great value to show through this clinical case the important value of the imaging modalities particularly the 18FDG. It is significant to remind to the scientific community the benefits of such kind of tools.

   **Answering:** Thank you very much for your approval!

2. Science editor:

   In figures' legends, indicate the type of the picture or the chart (ex: MRI picture of ...)

   - When you mention the laboratory investigation results, kindly accompany the result with the normal range or value.

   **Answering:** We have added the result with the normal range or value in figures' legends.